Author's response to reviews

Title: The Sydney Diabetes Prevention Program: A community-based translational study

Authors:

Stephen Colagiuri (Stephen.colagiuri@sydney.edu.au)
Philip Vita (Philip.vita@sydney.edu.au)
Magnolia Cardona-Morrell (magnolia.cardona-morrell@sydney.edu.au)
Maria Fiatarone-Singh (maria.fiataronesingh@sydney.edu.au)
Louise Farrell (farrelll@health.usyd.edu.au)
Andrew Milat (amiladoh.health.nsw.gov.au)
marion Haas (Marion.Haas@uts.edu.au)
Adrian Bauman (adrianb@health.usyd.edu.au)

Version: 2 Date: 25 May 2010

Author's response to reviews: see over
May 25, 2010

Dr Jigisha Patel, MRCP PhD.
Medical Editor
BMC – Public Health

Dear Dr Patel,

Re: BMC Public Health, MS: 8627162253746160
The Sydney Diabetes Prevention Program: A community-based translational study

In response to your email of May 19, I wish to confirm that the following edits have been made:

- **Page 1** The title page has now been included which includes the Title, Author list, Affiliations, and E-mail addresses
- **Page 2/3** Abstract inserted
- **Page 15** Indentation last line
- **Page 17** Abbreviations have been formatted into one paragraph
- **Page 28** Visible vertical lines have been removed from the Table 1
- We confirm that the manuscript has been checked for typographical errors.

Regards,

Stephen Colagiuri
Professor of Metabolic Health, Boden Institute of Obesity, Nutrition & Exercise